Author's response to reviews

Title: Role of Vasopressin in the Treatment of Anaphylactic Shock in a Child Undergoing Surgery for Congenital Heart Disease: a Case Report

Authors:

Luca Di Chiara (dichiaraluca@libero.it)
Giulia V Stazi (giuliavaleria@tiscali.it)
Zaccaria Ricci (zaccaria.ricci@fastwebnet.it)
Angelo Polito (angpolito@hotmail.com)
Stefano Morelli (s.zeus@inwind.it)
Chiara Giorni (c_giorni@yahoo.it)
Ondina La Salvia (dichiaraluca@libero.it)
Vincenzo Vitale (ezio.vitale@tin.it)
Eugenio Rossi (zaccaria.ricci@fastwebnet.it)
Sergio Picardo (picardo@opbg.net)

Version: 5 Date: 1 February 2008

Author's response to reviews:

Dear editor,

we made the required changes to the format of our manuscript so that online production process of your manuscript can proceed.

thanks for acceptance

best regards

zaccaria ricci